Author’s response to reviews

Title: Developing research priorities for palliative care of people with intellectual disabilities in Europe: a consultation process using nominal group technique

Authors:

Irene Tuffrey-Wijne (I.Tuffrey-Wijne@sgul.kingston.ac.uk)
Monika Wicki (monika.wicki@hfh.ch)
Pauline Heslop (Pauline.Heslop@bristol.ac.uk)
Mary McCarron (mccarrm@tcd.ie)
Stuart Todd (stuart.todd@southwales.ac.uk)
David Oliver (drdjoliver@gmail.com)
Anke de Veer (A.deVeer@nivel.nl)
Gerd Ahlström (gerd.ahlstrom@med.lu.se)
Sabine Schaeper (s.schaeper@katho-nrw.de)
Geralyn Hynes (hynesge@tcd.ie)
Janet O’Farrell (ofarrejc@tcd.ie)
Judith Adler (Judith.adler@hfh.ch)
Florian Riese (florianriese@gmail.com)
Leopold Curfs (leopold.curfs@maastrichtuniversity.nl)

Version: 1 Date: 16 Dec 2015

Author’s response to reviews:

We have now included the following sentences in the “Methods” section (line 210):
“Due to its methodology, the process did not fall under the Swiss Federal Act on Research on Human Beings and was therefore not subject to ethics review by the cantonal ethics committee in Zurich (where the workshop was held). All participants provided consent for study participation and publication or their views.”

In the author contributions, we have now used initials instead of authors’ full surnames.